Author’s response to reviews

Title: Temporal trends in incidence, recurrence and prevalence of stroke in an era of ageing populations. A longitudinal study of the total Swedish population.

Authors:

Karin Modig (karin.modig@ki.se)
Mats Talbäck (mats.talback@ki.se)
Louise Ziegler (louise.dencker-ziegler@sll.se)
Anders Ahlbom (anders.ahlbom@ki.se)

Version: 2 Date: 24 Jan 2019

Author’s response to reviews:

We have addressed all the comments raised by the Editor.